Author’s response to reviews

Title: Lessons from a Training Needs Assessment to strengthen the capacity of Routine Immunization Service Providers in Nigeria

Authors:
Linda Arogundade (lindaomosefe@yahoo.com; linda.arogundade@dclnigeria.com)
Titilola Akinwumi (titi.akinwumi@dclnigeria.com; titimakay@gmail.com)
Ebubechi Nwaononiuwu (Ebubechi.Nwaononiuwu@gmail.com; Ebubechi.Nwaononiuwu@dclnigeria.com)
Joshua Ezika (joshykime@gmail.com; joshua.ezika@dclnigeria.com)
Inuwa Yau (inuwabarau@yahoo.com)
Shola Molemodile (sholamole@gmail.com; shola.mole@dclnigeria.com)
Chizoba Wonodi (chizobabw@gmail.com)

Version: 7 Date: 02 Sep 2019

Author’s response to reviews:

Editor Comments:
1. Ethics and consent to participate

Please provide the full names of every ethics committee that provided ethics approval for this study in the ethics and consent to participate section. Please also include the full names of every committee that provided administrative approval for this study in this section.

Response
a) The ethics committee that provided approval for this study is the National Health Research Ethics Committee of Nigeria (NHREC). The respective protocol approval number are
   o NHREC Protocol Number NHREC/01/01/2007-14/07/2016 and
   o NHREC Approval Number NHREC/01/01/2007-17/05/2017

b) The respective agencies and institutions in the states where the study was done, gave administrative approval since an approval from NHREC that suffices for all institutions had already been obtained.
c) The names of the respective agencies and institutions involved in the study are:

- National Primary Health Care Development Agency, Abuja
- Bauchi State Primary Health Care Development Agency
- School of Health Technology Ningi, Bauchi
- College of Nursing and Midwifery, Bauchi state
- School of Nursing Abubakar Tafawa Balewa University Teaching Hospital (ATBUTH)
- College of Medical Sciences, Abubakar Tafawa Balewa University (ATBU), Bauchi state.
- Niger State Primary Health Care Development Agency
- School of Health Technology, Tungan Magajiya, Bauchi state
- College of Nursing, Bida; Niger state
- School of Midwifery, Minna Niger state
- School of Health Technology, Minna, Niger State
- Rivers State Primary Health Care,
- College of Health Science, University of Port Harcourt Teaching Hospital (CHS-UPTH), Rivers state
- College of Health Science and Technology (CHST); Rivers state
- School of Midwifery (SOM); and School of Nursing (SON), Rivers state

2. Declaration

Please remove the declaration section

"Declaration"

This is to certify that all authors have seen and approve to the final version of the manuscript being submitted. This also verifies that the article is the authors original piece of work and hasn’t been considered or received for prior publication elsewhere."

You can include this statement in the Competing interests’ section.

Response
This section has been moved it to Competing interests’ section.

3. Cite table 5 and supplementary files

We note that table 5 and the supplementary files have not been cited in the main text of the manuscript. Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

Response

Table 5 and supplemental files have been cited in the appropriate paragraphs in the manuscript

4. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Please remove any files that should not be published along with your manuscript e.g. cover letters, reviewer responses, guidelines etc., ready for publication.

Response

Done